Author’s response to reviews

Title: Effects of Stress on Behavior and Resting-State fMRI in Rats and Evaluation of Telmisartan Therapy in a Stress-Induced Depression Model

Authors:

Li junling (julian1224@126.com)
Yang ran (sunnygirl-1986@163.com)
xia kai (xiakai@vip.163.com)
Wang tian (shui_miao@qq.com)
nie binbin (niebb@ihep.ac.cn)
Gao kuo (515123134@qq.com)
chen jianxin (31314165@qq.com)
zhao huihui (hh686@126.com)
Li Yubo (15811221143@126.com)
Wei Wang (1614502448@qq.com)

Version: 4 Date: 04 Sep 2018

Author’s response to reviews:

Dear Dr. Matthew Hickey,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript and consideration on publication of our paper, we appreciate you and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “Effects of Stress on Behavior and Resting-State fMRI in Rats and Evaluation of Telmisartan Therapy in a Stress-Induced Depression Model”. (Manuscript No: BPSY-D-17-00717).

We have updated our Methods section to accurately reflect our method of sacrifice for rats euthanized in the course of our study (line 1-2, page 6). We have also checked our manuscript to
make sure it’s consistent with the format requirements. If there are still questions, please not hesitate to contact us. We will try our best to satisfy your requirements.

Again, we would like to express our great appreciation for giving us the opportunity. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Junling Li & Wei Wang